Author's response to reviews

Title: A randomized controlled trial of an exercise intervention targeting cardiovascular and metabolic risk factors for prostate cancer patients from the RADAR trial

Authors:

Daniel A Galvao (d.galvao@ecu.edu.au)
Nigel Spry (Nigel.Spry@health.wa.gov.au)
Dennis R Taaffe (dtaaffe@hms.uq.edu.au)
James Denham (Jim.Denham@newcastle.edu.au)
David Joseph (David.Joseph@health.wa.gov.au)
David S Lamb (David.Lamb@Otago.ac.nz)
Greg Levin (g.levin@ecu.edu.au)
Gillian Duchesne (Gillian.Duchesne@petermac.org)
Robert U Newton (r.newton@ecu.edu.au)

Version: 2 Date: 3 November 2009

Author's response to reviews: see over
BMC Cancer

November 2, 2009

Dear Professor Melissa Norton

Thank you for reviewing our manuscript #1724911507306613 entitled “A randomized controlled trial of an exercise intervention targeting cardiovascular and metabolic risk factors for prostate cancer patients from the RADAR trial”, and giving us a chance to revise it.

We believe that we have addressed the few additional requested changes.

We trust this manuscript is now suitable for publication in BMC Cancer.

Thank you for your consideration.

Sincerely,

Daniel A. Galvão, PhD
Senior Research Fellow
School of Exercise, Biomedical and Health Sciences
Edith Cowan University, Australia

Details of revisions:

MS # 1724911507306613: A randomized controlled trial of an exercise intervention targeting cardiovascular and metabolic risk factors for prostate cancer patients from the RADAR trial

1 - Please document that you will seek informed consent from participants of the trial in the Methods section of your manuscript.

The requested information has been added to the following sentence on page 9 of the revised manuscript and now reads:

“The protocol has been approved (ID: 3636-GALVÃO) by the University Human Research Ethics Committee and all subjects will provide written informed consent.”

2 - Please include your Trial Registration number at the bottom of your Abstract.

Trial number has been added at the bottom of the abstract.

3 - Please ensure that the sections of your manuscript are in the right order.

Sections of the manuscript have been revised and are in the right order. Figure Legends have been added after the References as per guidelines.